Author's response to reviews

**Title:** Modeling repeated ordinal responses using a family of power transformations: application to neonatal hypothermia data

**Authors:**

Farid Zayeri (fzayeri@yahoo.com)
Anoshirvan Kazemnejad (aklili@yahoo.com)
Navid Khanafshar (navka@yahoo.com)
Fatemeh Nayeri (fnayeri@sina.tums.ac.ir)

**Version:** 6  **Date:** 4 September 2005

**Author’s response to reviews:** see over
Dear Editorial Board

Hereby, I wish to express thanks the editors of BMC Medical Research Methodology for approving the publication of our manuscript entitled "Modeling repeated ordinal responses using a family of power transformations: application to neonatal hypothermia data".

I would like to inform you that we changed the manuscript according to your requested revisions. Please see the following descriptions.

If there is anything else I have to do it, please do not hesitate to ask.

Yours truly,
Farid Zayeri.

Department of Biostatistics,
School of Medical Sciences,
Tarbiat Modarres University,
P.O.Box: 14115-111, Tehran, Iran.
Phone: #98-21-8013030, Fax: #98-21-8006544.
E-mail: fzayeri@yahoo.com
Revisions

1. The authors' titles and qualifications were removed from the title page.

2. The affiliations were provided in the requested format.

3. The authors were linked to their affiliations according to the described format.

4. The *keywords* section was removed from the manuscript.

5. The order of the manuscript sections was changed according the illustrated format.

6. The 'Competing interests' section was provided.

7. The 'Acknowledgements' section was provided.

8. The 'Appendix" section was placed before the 'Competing interests' section.